Author’s response to reviews

Title: Treatment outcomes in patients with opioid use disorder initiated by prescription: A systematic review protocol

Authors:

Nitika Sanger (sangern@mcmaster.ca)
Meha Bhatt (bhattm25@mcmaster.ca)
Laura Zielinski (zielinsl@mcmaster.ca)
Stephanie Sanger (sangers@mcmaster.ca)
Hamnah Shahid (shahih4@mcmaster.ca)
Bianca Bantoto (bantotb@mcmaster.ca)
M.Constantine Samaan (samaanc@mcmaster.ca)
Russell de Souza (rdesouz@mcmaster.ca)
Zainab Samaan (samaanz@mcmaster.ca)

Version: 2 Date: 18 Dec 2017

Author’s response to reviews:

SYSR-D-17-00131

Treatment outcomes in patients with opioid use disorder initiated by prescription: A systematic review protocol

Nitika Sanger, M.Sc (c); Meha Bhatt; Laura Zielinski; Stephanie Sanger; Hamnah Shahid; Bianca Bantoto; M.Constantine Samaan; Russell de Souza; Zainab Samaan
Response to reviewers’ comments

We have the pleasure in resubmitting a revised manuscript addressing the editorial comments. We would like to thank the editors for considering an improved version of the manuscript and the reviewers for the helpful critique and valuable comments.

Reviewer reports:

Thank you for re-submitting your manuscript with responses to the reviewer and editor queries. All the points raised have been adequately addressed and the manuscript reads well. There is just a very minor edit required of two typos:

p10 line 243 'consists' should read 'consist'

Author’s Response: Thank you for pointing this out. This has been fixed. Please see page 10 Line 243.

p11 line 273 'separately as. pooling' should not contain a full stop.

Author’s Response: Thank you for pointing this out. This has been fixed. Please see page 11 Line 273.